Author’s response to reviews

Title: The Clinical Efficacy of Afatinib 30 mg Daily As Starting Dose May Not Be Inferior to Afatinib 40 mg Daily in Patients with Stage IV Lung Adenocarcinoma Harboring Exon 19 or Exon 21 Mutation

Authors:

chih-jen yang (chjeya@cc.kmu.edu.tw)
Ming-Ju Tsai (960216kmuh@gmail.com)
Jen-Yu Hung (jenyuhung@gmail.com)
Yu-Chen Tsai (poppy_snow@hotmail.com)
Jui-Feng Hsu (940360@mail.kmuh.org.tw)
Ta-Chih Liu (d730093@cc.kmu.edu.tw)
Ming-Shyan Huang (shyang@kmu.edu.tw)
Inn-Wen Chong (chong@kmu.edu.tw)
Ying-Ming Tsai (yingming@kmu.edu.tw)
Mei-Hsuan Lee (mhsuan99@yahoo.com.tw)

Version: 3 Date: 31 Jul 2017

Author’s response to reviews:

Dear Editors and Reviewers:

Thank you for giving us the opportunity to improve our manuscript. We have revised the manuscript based on the suggestions from the reviewers. We have also answered the questions raised by the reviewers in a point-by-point fashion below.
1. We notice the order of authors provided on the title page is different from the order listed in the submission data. Please confirm the correct order of authorship and provide the correct order in both the submission system and on the title page.

Reply. I had corrected based on your kind comments.

2. Please include a Conclusions heading in the text of your manuscript, as both a Conclusion and a Discussion section should be provided. You may consider placing it before the last paragraph of the Discussion section.

Reply. I had corrected based on your kind comments.

3. Please edit your Acknowledgements section, as this information is already provided under "Funding." In the Acknowledgements, please acknowledge anyone who contributed towards the article who does not meet the criteria for authorship including anyone who provided professional writing services or materials. If this does not apply to your manuscript, then please state "Not applicable."

Reply. I had corrected based on your kind comments.

4. Please add a section "Consent for publication" in the Declarations. You may state "Not applicable" in this section as no personal information is provided in your manuscript.

Reply. I had corrected based on your kind comments.

5. In the Authors' contributions section, please refer to all authors by their initials. For example, CJ Yang should be "CJY."

Reply. I had corrected based on your kind comments.
6. Please provide more justification for the contributions of author CJ Yang, as currently, their contributions do not automatically qualify for authorship. Acquisition of funding, collection of data, or general supervision of the research group, alone, does not usually justify authorship, so please amend your manuscript accordingly.

Reply. I had corrected based on your kind comments.